Author's response to reviews

Title: Mediators of the relation between War Experiences and Suicidal Ideation among Former Child Soldiers in Northern Uganda: The WAYS Study

Authors:

Kennedy Amone-P’Olak (kpamone@gmail.com)
Tlholego M Lekhutlile (Tlholego.Lekhutlile@mopipi.ub.bw)
Emilio Ovuga (emilio.ovuga@gmail.com)

Version: 2 Date: 8 April 2014

Author's response to reviews:

Details of revisions:

1. The manuscript has been revised and the line and page numbers included.

2. Statement of competing interest has been included appropriately as suggested by the editor.

3. Authors’ contributions have been included before the acknowledgement section and reference list.

4. The acknowledgement has been included between the Authors’ contributions section and Reference list as suggested by the editor.

5. a) Permission from all those mentioned in the acknowledgement was sought.

b) Source of funding for the study has been indicated appropriately within the acknowledgement section.

6. Ethics statement has been included in the methods section as suggested.